**Author’s response to reviews**

**Title:** A Novel Pathogenic Variant in OSBPL2 Linked to Hereditary Late-onset Deafness in a Mongolian Family

**Authors:**

Ningjin Wu (ninjinuria@163.com)

Husile Husile (husile320000@163.com)

Liqing Yang (yangliqing1991@163.com)

Yaning Cao (ya_ning_cao@163.com)

Xing Li (452976626@qq.com)

Wenyan Huo (wenyan_555@163.com)

Haihua Bai (hh_bai@163.com)

Yangjian Liu (yangjianliu@gmail.com)

Qizhu Wu (qizhu_wu2014@163.com)

**Version:** 3  **Date:** 12 Feb 2019

**Author’s response to reviews:**

February 13, 2019

BMC Medical Genetics

Dear editor,

Thank you to the editor for accepting our article.

According to the editor's suggestions, we revised the manuscript “A Novel Pathogenic Variant in OSBPL2 Linked to Hereditary Late-onset Deafness in a Mongolian Family” (MGTC-D-18-00356R3), and the final version has been submitted.
Thanks again!

Sincerely,

Qizhu Wu, Prof.

Affiliated Hospital of Inner Mongolia University for Nationalities
No.1742 Huolinhe Street, Keerqin District, Tongliao City, Inner Mongolia
028007, China.

Email: qizhu_wu2014@163.com
Tel: 86 475 8435422